Author’s response to reviews

Title: The intellectual structure and substance of the knowledge utilization field: A longitudinal author co-citation analysis, 1945-2004

Authors:

Carole A. Estabrooks (carole.estabrooks@ualberta.ca)
Linda Derksen (derksenl@mala.ca)
Connie Winther (connie.winther@ualberta.ca)
John N. Lavis (lavisj@mcmaster.ca)
Shannon D. Scott (shannon.scott@ualberta.ca)
Lars Wallin (lars.wallin@karolinska.se)
Joanne Profetto-McGrath (joanne.profetto-mcgrath@ualberta.ca)

Version: 2 Date: 16 July 2008

Author's response to reviews: see over
15 July 2008

Martin Eccles
Editor, Implementation Science
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB, UK

Dear Martin,

RE: The intellectual structure and substance of the knowledge utilization field: A longitudinal
author co-citation analysis, 1965-2004 (MS: 9116368171750532)

We have uploaded the above named manuscript which has been revised.

We have also uploaded our response to the three reviewers’ comments.

Thank you for your consideration of this manuscript. I look forward to your response.

Yours sincerely,

Carole A. Estabrooks
Professor & Canada Research Chair
in Knowledge Translation

cc Linda Derksen
John Lavis
Connie Winther
Shannon Scott
Lars Wallin
Joanne Profetto-McGrath